Author's response to reviews

Title: Spinal cord injury secondary to high-voltage electrocution: a case report.

Authors:

Harpreet K. Johl (harpreetjohl@gmail.com)
Adel Olshansky (adel.olshansky@gmail.com)
Said R. Beydoun (sbeydoun@usc.edu)
Richard A. Rison (rison@usc.edu)

Version: 2 Date: 28 February 2012

Author's response to reviews: see over
February 27th, 2012

Dear Professor Michael Kidd,

Submitted for your consideration is a paper entitled “Spinal cord injury secondary to high-voltage electrocution: a case report.” We present the case of a middle-aged gentleman who suffered a high-voltage electrocution injury resulting in cervical ischemic myelopathy and secondary paraparesis confirmed with neuroimaging studies. Given a relative paucity of published literature with regard to neurologic injury and treatment in this entity, we feel that this case report will hold a wide readership interest given the multiple medical professionals involved in the care of these patients, including trauma, burn, and plastic surgeons along with internists, family practitioners, hospitalists, and neurologists.

Please note that we have been able to obtain full consent now from the patient for full publication of this case report and the accompanying images. The consent section has therefore been appropriately reworded.

Sincerely yours,

Richard A. Rison, M.D., F.A.A.N.E.M.